Author's response to reviews

Title: Clinical predictors of lacunar syndrome not due to lacunar infarction

Authors:

Adrià Arboix (aarboix@hscom.com)
Joan Massons (jmassons@terra.es)
Luís Garcia-Eroles (lgarciaer@csdm.es)
Cecília Targa (ctargab@meditex.es)
Emili Comes (13528ecm@comb.cat)
Olga Parra (oparra@ub.edu)

Version: 4 Date: 13 May 2010

Author's response to reviews:

Melissa Norton, MD
Editor-in-Chief, BMC Neurology

Dear Dr Norton:

Through the http://www.biomedcentral.com/manuscript/bmcneurol we have uploaded the original revised manuscript entitled “Clinical predictors of lacunar syndrome not due to lacunar infarction” which is submitted to the consideration of the Editorial Board of BMC Neurology as an Original Research article.

We have included the formatting changes suggested in the letter dated May 12, 2010.

Thank you very much for your attention to this submission.

Yours sincerely,

Adrià Arboix, MD, PhD
Cerebrovascular Division, Department of Neurology
Hospital Universitari del Sagrat Cor
C/ Viladomat 288
E-08029 Barcelona, Spain
Telephone number +34–93–4948940; fax +34–93–4948906,
E-mail: aarboix@hscom.com